

### Denisovan introgression has shaped the immune system of present-day Papuans

Davide M Vespasiani, Guy S Jacobs, Laura E Cook, Nicolas Brucato, Matthew Leavesley, Christopher Kinipi, Francois-Xavier Ricaut, Murray P Cox, Irene Gallego Romero

### ► To cite this version:

Davide M Vespasiani, Guy S Jacobs, Laura E Cook, Nicolas Brucato, Matthew Leavesley, et al.. Denisovan introgression has shaped the immune system of present-day Papuans. PLoS Genetics, 2022, 18 (12), pp.e1010470. 10.1371/journal.pgen.1010470. hal-04222847v3

### HAL Id: hal-04222847 https://hal.science/hal-04222847v3

Submitted on 29 Sep 2023

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



### G OPEN ACCESS

**Citation:** Vespasiani DM, Jacobs GS, Cook LE, Brucato N, Leavesley M, Kinipi C, et al. (2022) Denisovan introgression has shaped the immune system of present-day Papuans. PLoS Genet 18(12): e1010470. https://doi.org/10.1371/journal. pgen.1010470

**Editor:** Anna Di Rienzo, University of Chicago, UNITED STATES

Received: February 1, 2022

Accepted: October 10, 2022

Published: December 8, 2022

**Peer Review History:** PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pgen.1010470

**Copyright:** © 2022 Vespasiani et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All requests to access the sequences from Jacobs et al are managed through the Data Access Committee of the official data repository (accession RESEARCH ARTICLE

# Denisovan introgression has shaped the immune system of present-day Papuans

Davide M. Vespasiani<sup>1,2</sup>, Guy S. Jacobs<sup>3</sup>, Laura E. Cook<sup>1,2</sup>, Nicolas Brucato<sup>4</sup>, Matthew Leavesley<sup>5,6,7</sup>, Christopher Kinipi<sup>5</sup>, François-Xavier Ricaut<sup>4</sup>, Murray P. Cox<sup>8</sup>, Irene Gallego Romero<sup>1,2,9,10</sup>\*

1 Melbourne Integrative Genomics, University of Melbourne, Parkville, Australia, 2 School of Biosciences, University of Melbourne, Parkville, Australia, 3 Department of Archaeology, University of Cambridge, Cambridge, Uniteed Kingdom, 4 Laboratoire de Evolution et Diversite Biologique, Université de Toulouse Midi-Pyrénées, Toulouse, France, 5 School of Humanities and Social Sciences, University of Papua New Guinea, Port Moresby, Papua New Guinea, 6 College of Arts, Society and Education, James Cook University, Cairns, Australia, 7 ARC Centre of Excellence for Australian Biodiversity and Heritage, University of Wollongong, Wollongong, Australia, 8 School of Natural Sciences, Massey University, Palmerston North, New Zealand, 9 Center for Stem Cell Systems, University of Melbourne, Parkville, Australia, 10 Center for Genomics, Evolution and Medicine, University of Tartu, Tartu, Estonia

\* irene.gallego@unimelb.edu.au

### Abstract

Modern humans have admixed with multiple archaic hominins. Papuans, in particular, owe up to 5% of their genome to Denisovans, a sister group to Neanderthals whose remains have only been identified in Siberia and Tibet. Unfortunately, the biological and evolutionary significance of these introgression events remain poorly understood. Here we investigate the function of both Denisovan and Neanderthal alleles characterised within a set of 56 genomes from Papuan individuals. By comparing the distribution of archaic and non-archaic variants we assess the consequences of archaic admixture across a multitude of different cell types and functional elements. We observe an enrichment of archaic alleles within cisregulatory elements and transcribed regions of the genome, with Denisovan variants strongly affecting elements active within immune-related cells. We identify 16,048 and 10,032 high-confidence Denisovan and Neanderthal variants that fall within annotated cisregulatory elements and with the potential to alter the affinity of multiple transcription factors to their cognate DNA motifs, highlighting a likely mechanism by which introgressed DNA can impact phenotypes. Lastly, we experimentally validate these predictions by testing the regulatory potential of five Denisovan variants segregating within Papuan individuals, and find that two are associated with a significant reduction of transcriptional activity in plasmid reporter assays. Together, these data provide support for a widespread contribution of archaic DNA in shaping the present levels of modern human genetic diversity, with different archaic ancestries potentially affecting multiple phenotypic traits within non-Africans.

EGAS00001003054) at the European Genomephenome Archive (EGA; <u>https://www.ebi.ac.uk/ega/</u> home). All remaining data are within the manuscript and its <u>Supporting information</u> files. All analyses were performed using R software v 4.0.0. The complete list of scripts used for the analyses is available at <u>https://gitlab.unimelb.edu.au/igr-lab/</u> Archaic\_introgression\_in\_PNG.

Funding: This work was supported by an award from the Leakey Foundation and by Australian Research Council Discovery Project DP200101552, both to I.G.R. and by The French National Research Agency (ANR) (grant PAPUAEVOL n° ANR-20-CE12-0003-01 (F.X.R). N.B. was supported by the PAPUAEVOL grant. D.M.V was supported by the University of Melbourne's Albert Shimmins Fund. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

#### Author summary

Humans of Papuan ancestry owe roughly 5% of their genome to Denisovans, a poorly characterised archaic hominin. While introgressed DNA segments can be readily identified, understanding their biological consequences remains challenging. By examining the distribution of introgressed DNA against existing functional genomics datasets, it is possible to predict the phenotypes they impact. In Papuans, Denisovan DNA, but not Neanderthal, strongly and consistently affects immune cells and immune-related processes of potential evolutionary relevance. In vitro testing of introgressed variants confirms these predictions, suggesting Denisovan variants can impact gene regulation in vivo. Variation in gene expression might be key to understanding the consequences of admixture between modern humans and archaic hominins, as has been observed with Neanderthal DNA in other human populations.

#### Introduction

Modern humans are known to have interbred with Neanderthals [1], Denisovans [2] and possibly other archaic hominins [3]. While genetically similar populations of Neanderthals are thought to have contributed approximately 2% to non-African genomes, Denisovan introgression has been observed to be more variable [4]. Particularly, Denisovan ancestry accounts for up to 5% of the genomes of the Indigenous peoples of Island Southeast Asia and Australia [4, 5]. In addition, these components exhibit a deep divergence from the reference Altai Denisovan genome, providing strong evidence for the occurrence of multiple Denisovan introgression events across time and space [6, 7].

At the genomic level these introgressed archaic alleles are mostly observed outside proteincoding sequences, distributed over non-functional and regulatory regions [8, 9]. Enhancers, in particular, are amongst the top targeted elements of archaic hominin introgression [10, 11]. Here, archaic alleles are thought to drive phenotypic differences by altering gene pre- and post-transcriptional regulatory processes [11]. Furthermore, both Neanderthal and Denisovan variants seem to preferentially affect enhancers in a tissue-specific manner, with highly pleiotropic elements being depleted of archaic variation [12]. However, beyond general agreement that introgressed archaic DNA has mainly been deleterious and actively removed from coding sequences and conserved non-coding elements [8, 12–14], the actual phenotypic consequences of this variation are not well understood. Several lines of evidence highlight associations between archaic DNA and risk for disease traits, including autoimmune diseases [8, 15, 16], or with traits of possible evolutionary advantage for early non-Africans [17–20]. For example, Neanderthal variants within immune genes and immune-related cis-regulatory elements (CREs) have been associated with differential responses to viral infections among present-day Europeans [21, 22].

Unfortunately, two main factors limit our interpretation of the biology of these alleles. First, we still lack a detailed characterisation of global levels of human genetic diversity, vital to identifying differential archaic hominin contributions across modern populations [23, 24]. Because Denisovan DNA is not present in the genomes of modern Europeans, which make up the vast diversity of biomedical genetics cohorts, it has been impossible to link them to phenotype, unlike segregating Neanderthal variants [16, 25]. Second, and more challenging, most of these alleles lie within non-coding sequences where, despite their acknowledged contributions to human evolutionary history [26, 27], an understanding of their actual biological functions remains elusive. To gain insights into the consequences of archaic introgression in Papuans, we have analysed a previously published dataset of 56 present-day genomes sampled across the island of New Guinea [6]. By comparing the distribution of archaic single nucleotide polymorphisms (aSNPs) and non-archaic SNPs (naSNPs) segregating within these populations, across multiple genomic elements and cell types, we find that aSNPs are enriched within functional cis-regulatory elements (CREs) and transcribed regions, particularly those active within immune-related cells. We also observe that the presence of archaic alleles within these elements can lead to substantial disruption of transcription factor binding sites (TFBSs), with a consistent Denisovanspecific signal affecting immune-related processes. We finally validate these results through experimental testing of multiple Denisovan aSNPs, and find they are associated with significant transcriptional changes in reporter gene experiments.

#### Materials and methods

#### **Ethics statement**

All human genome data in this study was taken from Jacobs *et al.* [6]. All collections followed protocols for the protection of human subjects established by institutional review boards at the Eijkman Institute (EIREC #90) and Nanyang Technological University (IRB-2014–12-011); the analyses in this publication were additionally approved by University of Melbourne's Human Ethics Advisory Group (1851585.1). Permission to conduct research in Indonesia was granted by the State Ministry of Research and Technology (RISTEK). All individuals gave their full informed written consent to participate in the study.

The two lymphoblastoid cell lines (LCLs) used in the plasmid reporter experiments were established from donors sampled at the University of Papua New Guinea, in Port Moresby, Papua New Guinea. Collection of material from healthy adult donors was coordinated by Dr Christopher Kinipi (Director of Health Services at the University of Papua New Guinea) and approved by the Medical Research Advisory Committee of the National Department of Health of the Government of Papua New Guinea (permit number MRAC 16.21), by the University of Melbourne (ID: 1851585.1), and by the French Ethics Committees (Committees of Protection of Persons 25/21\_3, n°SI:21.01.21.42754). Permission to conduct research in Papua New Guinea was granted by the National Research Institute of Papua New Guinea (permit 99902292358), with full support from the School of Humanities and Social Sciences, University of Papua New Guinea. These approvals commit our team to following all ethical guidelines mandated by the government of Papua New Guinea. All individuals gave their full informed written consent to participate in the study.

#### Variant filtering and selection

A list of 7,791,042 SNPs segregating within non-archaic, Denisovan and Neanderthal haplotypes identified in 56 genetically Papuan individuals was taken from [6] (for detailed information about samples, genotyping and inference of introgressed state, see [6]). To define a high confidence set of introgressed SNPs and a matched set of SNPs found outside archaic haplotypes to use as controls, we applied a series of stringent filtering steps to the genome-wide data, as follows. First, all singletons were removed, together with variants lacking ancestral state information in the 6 non-human primate EPO alignment [28]. Variants were then labelled as possibly archaic if they fell within archaic haplotypes defined by [6], and further classified as Denisovan or Neanderthal aSNPs on the basis of the inferred ancestry of the haplotype. Due to incomplete lineage sorting and the relatively short divergence time between Denisovan and Neanderthal lineages [2], a single archaic SNP can segregate in both Neanderthal and Denisovan haplotypes. We observed 18,518 aSNPs that segregated in both Denisovan and Neanderthal haplotypes. These variants were assigned to one of the two archaic ancestries first by considering in which archaic haplotype they segregated at higher frequencies and, then, by comparing the state of the main introgressed allele in Papuans to the Altai Neanderthal [29] or the Altai Denisovan [2] reference genomes. This resulted in 6,098 and 9,744 variants that were respectively assigned to Denisovan and Neanderthal ancestries. The 2,676 variants that could not be disambiguated in this manner were labelled as ambiguous and excluded from downstream analyses.

To further control for incomplete lineage sorting between humans, Neanderthal and Denisovans, and possible errors in haplotype calling, allele frequencies were calculated by considering the number of archaic or non-archaic haplotypes carrying the allele of interest relative to the total number of chromosomes in the sample. Specifically, for naSNPs we calculated the derived allele frequency (DAF) as the number of derived alleles (relative to the EPO alignment) segregating within non-archaic haplotypes divided by the total number of chromosomes genotyped at that site. Likewise, for each aSNP, we calculated the main introgressed allele frequency (MIAF) by counting the number of observations of the most abundant allele found within archaic haplotypes divided by the total number of chromosomes genotyped at that site. Variants with MIAF or DAF < 0.05 or  $\ge 0.05$  were respectively labelled as low-frequency or common to high-frequency. We also calculated the frequency of the main introgressed allele within non-archaic haplotypes for all aSNPs, and then examined the differences in frequency across modern and archaic haplotype backgrounds. This showed that in a large number of cases, alleles that segregate within archaic haplotypes also segregate within individuals that carry a non-archaic haplotype at the same site. To address this confounder, we calculate allele frequencies in archaic and non-archaic haplotypes, and removed all SNPs where the difference in frequency between archaic and non-archaic haplotypes was < 0.25 (S1 Fig). This step does also exclude some aSNPs segregating at overall low frequencies in our samples and thus reduces the overall size of our aSNP sets, but leads to a higher-confidence dataset. Finally, we removed all aSNPs and naSNPs for which the main introgressed allele or derived allele, respectively, segregated at frequencies > 0.005 within the sub-Saharan African populations in the 1000 Genomes Project [30].

Following these steps, we retained a set of 140,916, 88,625 and 1,108,747 variants of putative Denisovan, Neanderthal and non-archaic ancestry, respectively. A detailed overview of the number of variants retained after each filtering step can be found at S1 Table. Inspection of the site frequency spectrum of these three sets (SFS) revealed a different distribution of the allele frequencies between aSNPs and naSNPs (S2(A) and S2(B) Fig). To account for this, we randomly sub-sampled a total of 229,541 naSNPs (corresponding to the combined number of aSNPs) from the set of non-archaic variants to match the SFS of both Denisovan and Neander-thal aSNPs and used this set for all downstream comparisons (S2(C) Fig). The final sets of Denisovan, Neanderthal and non-archaic variants are listed in S2–S4 Tables.

# Definition of a control set of Neanderthal variants segregating within West Eurasians

We additionally defined a control set of Neanderthal variants following the same principles as above. Briefly, Neanderthal haplotype inferences were taken from 75 West Eurasian individuals from the Simons Genome Diversity Project [31] that were analysed in [6], retrieving the Neanderthal haplotypes inferred and extracting allele counts for each SNP in Neanderthal and non-Neanderthal inferred haplotypes. Filtering steps to obtain a high-confidence set of Neanderthal variants segregating in West Eurasians were performed as above.

#### Quantifying the levels of background selection

To ensure the observed differences between aSNPs and naSNPs were not caused by different levels of background selection we downloaded from https://github.com/gmcvicker/bkgd the bed file containing the list of pre-computed values for the B-statistic in hg19 coordinates [32]. We then intersected these regions with the list of aSNPs and naSNPs to test whether the distribution of the corresponding values were comparable across the three ancestries. As a further control, we also calculated the distribution of the B-values for all know human protein-coding sequences (S3(A) and S3(B) Fig).

#### **Functional annotation**

To functionally annotate the refined set of SNPs we first used the annotatr R package [33] to determine the distribution of aSNPs across multiple genomic elements and whether they were significantly depleted from protein-coding sequences. To investigate the function of aSNPs across chromatin states and human cell types, we downloaded the 15-state hg19 mnemonics BED files containing the segmented chromatin state predictions for each of the 111 cell types included in the Roadmap Epigenomics project [34], https://egg2.wustl.edu/roadmap/web\_ portal/index.html. Variants were intersected with chromatin state files, and the 111 cell types were grouped into the 18 tissues as in [34]. We counted the number of cell types across which each SNP was annotated within the same chromatin state as a measure of its potential pleiotropic activity. Finally, we downloaded the list of V8 cis-eQTLs and their eGenes from the GTEx web portal (https://www.gtexportal.org/home/) [35], retaining only the positional information for the list of significant cis-eQTLs (i.e., qval < 0.05). Variants associated with multiple genes were counted only once. The genomic coordinates for the resulting list of cis-eQTLs were then lifted over from hg38 to hg19 using liftOver from the rtracklayer R package [36]. We used this dataset to identify shared instances first only with our set of common-to-high-frequency variants annotated within putative CREs and then with our entire set of 140,916 Denisovan, 88,625 Neanderthal and 229,541 non-archaic variants. In both analyses, we retained only those variants where the reference and alternative alleles matched between the datasets.

# Computing $F_{st}$ values between western Indonesia and New Guinea populations

 $F_{st}$  values were calculated between western Indonesian populations and New Guinean populations using the BCF files from [6] and the VCFtools *–weir-fst-pop* command. The western Indonesian population set was defined as in [6].

#### Enrichment of aSNPs across chromatin states and tissues

To quantify the impact of aSNPs within each chromatin state and/or across each chromatin state/tissue combination, we counted the number of Denisovan, Neanderthal and non-archaic variants annotated in each combination of interest. We then computed the odds ratio (OR) between the number of aSNPs and naSNPs for Denisovan and Neanderthal variants. Statistically significant deviations from an OR = 1 were tested using Fisher's exact test. Unless otherwise noted, resulting p-values were corrected for multiple hypothesis testing using the Benjamini-Hochberg (BH) method [37]. Enrichment scores together with their corresponding p-values for each comparison of interest can be found at S5–S7 Tables.

#### Impact of aSNPs on transcription factor binding sites

To quantify the impact of archaic alleles on transcription factor binding sites (TFBSs), we focused on variants annotated to TssA, TssAFlnk, TxFlnk, Enh and EnhG in at least one cell type. We then used the motifbreakR R package [38] to identify putative TFBS-disrupting variants. Position weight matrices (PWMs) were retrieved from Jaspar 2018 [39] and HOCO-MOCO v.11 [40]. Motifs were matched to the hg19 genome using p-value threshold of  $1^{-5}$ , and the allelic impact on each PWM (PWM score) was quantified via the built-in sum of log-probabilities algorithm specifying a background probability of 0.3 and 0.2 for A/T and G/C, respectively. Differences in PWM score between the two alleles at a site, defined as  $\Delta$  PWM, were polarised by the introgressed or derived state of aSNPs and naSNPs, respectively. Differences in PWM score between the two alleles at a site, defined as  $\Delta$  PWM, were polarised by the introgressed or derived state of aSNPs and naSNPs, respectively. Differences in PWM score between the two alleles as  $\Delta$  PWM, were polarised by the SNP-PWM associated with the same TF across both databases we retained only the SNP-PWM association with the highest absolute  $\Delta$  PWM score.

To assess the genome-wide impact of aSNPs on TFBSs while avoiding redundancy between TFs due to similar binding motif preferences, we grouped all TFs into families based on motif clustering information from [41]. We removed 67 HOCOMOCO and 62 Jaspar PWMs that were either not associated with any cluster or that have mismatches in the PWM versions between MotifDb [42] and [41]. We then tested whether any motif cluster contained an excess of TFBS-disrupting aSNPs compared to naSNPs by performing a  $\chi^2$ -test. Differences in  $\Delta$  PWM score distributions across ancestries were tested via a Wilcoxon test.

#### Gene ontology enrichment analysis

To understand the biological processes potentially affected by Denisovan and Neanderthal introgression, we retrieved all CRE-associated variants annotated in at least one haematopoetic stem cell (HSC) & B and/or blood & T related cell type. We then used the rGREAT R package package [43] to assign each variant to the regulatory domain of the nearest gene located within 1 Mb of distance and to perform a gene ontology (GO) enrichment analysis [44], using as a background set the union of all the TFBS-disrupting variants associated genes (both archaic and non-archaic). Importantly, by linking variants to the regulatory domain of its nearest gene, this approach does not yield multiple SNP-genes associations for a single SNP which could occur when in presence of gene clusters. Significantly enriched GO terms were defined as those with BH-corrected p-value < 0.01 and can be found in <u>S8 Table</u>.

# Quantifying the impact of Denisovan alleles on gene expression using a plasmid reporter assay

For each of the 5 selected SNPs (see Results section), we designed 2 DNA oligonucleotides of 170 nucleotides (nt) centred on the two alleles of interest (84 bp on the 5' and 85 bp on the 3' end). To ensure the effectiveness and reproducibility of our design, we also included a positive control variant (rs9283753) from Tewhey *et al.* 2016 [45]. All variants were synthesised on the positive strand, regardless of variant orientation with respect to its predicted target gene, as enhancer activity in reporter assays has been found to be largely independent of orientation [46]. In addition, all oligos contained two common 15 bp adapter sequences flanking each regulatory element, a 44 bp sequence centred on a BsiWI restriction site, a oligo-specific 15 bp barcode sequence and a universal 20 bp buffer sequence, resulting in a final oligo length of 284 bp. Oligos were synthesised as a pool by IDT. A list with the full sequences and DNA primers

used to amplify the oligo library (see below) is available at <u>S9 Table</u>. The pMPRA1 (Addgene, 49349) was chosen as plasmid backbone for the molecular cloning.

Synthesised oligos underwent two consecutive runs of Q5 (NEB, M0491) low-cycle PCR using two different sets of primers, first to make them double-stranded and then to add a SfiI restriction site at each end of the sequence. After each PCR amplification, oligos were purified with AMPure XP beads (Beckman Coulter, A63881) following the manufacturer's protocol. Amplified oligos and pMPRA1 vector were both SfiI digested (NEB, R0123) for an hour. Digestion solutions were then ran on an agarose gel, extracted using the gel extraction kit (NEB, T1020) and ligated using T4 DNA Ligase (NEB, M0202), again following the corresponding manufacturer's protocol. Ligated products were then transformed into E.coli competent cells (Promega, JM109), which were grown overnight in 25 mL of LB media supplemented with 100  $\mu$ g/mL of ampicillin (Thermo Fisher Scientific, FSBBP1760–5) prior to plasmid purification (Zymo, D4209). To create the final library, assembled plasmids were linearised using the BsiWI-HF (NEB, R3553) restriction enzyme. An amplicon containing a minimal promoter, GFP open reading frame (ORF), synthesised as a gblock by IDT (S9 Table), was then inserted by Gibson assembly (NEB, E2621). The resulting ligated products were again transformed into bacterial cells, which were overnight grown before subsequent plasmid extraction (Zymo, D4200). For each step, the transformation efficiency was estimated to be  $> 10^6$  cfu. We ensured correct assembly of each of the DNA oligos by Sanger sequencing (Macrogen).

The two LCLs used for experimental validation (PNG15 and PNG21) were grown in RPMI (Lonza, 12-702Q) supplemented with 1% Glutamax, 1% NEAA and 10% FBS. Cells were maintained at a cell density of  $5 \cdot 10^5$  cells/mL. Each cell line was electroporated in triplicate using the Neon transfection system (Thermo Fisher Scientific, MPK5000). For each transfection,  $1 \cdot 10^6$  cells were centrifuged at 700x g and resuspended in 100  $\mu$ l of R buffer with 1  $\mu$ g of plasmid library before applying 3 pulses of 1200 V for 20 ms each [45]. After transfection, cells were allowed to recover in 2 mL of culture media for 24 hours then collected by centrifugation and frozen at -80°C. For each transfection replicate, total DNA, including plasmid DNA (pDNA), and RNA were extracted from cells using Qiagen AllPrep kit (Qiagen, 80204) following the manufacturer's protocol, including on-column DNase digestion of the RNA fraction. pDNA and total RNA samples were quantified using the Qbit broad range dsDNA (Thermo Fisher Scientific, Q32850) and RNA kits (Thermo Fisher Scientific, Q10211), respectively. First-strand cDNA was synthesised from 1  $\mu$ g of mRNA for each sample using the High-Capacity cDNA Reverse Transcription Kit (Applied Biosystems, 4368814) and an universal custom reverse primers binding downstream to the oligo-specific barcode sequence (see S9 Table).

The regulatory activity of each oligo was then quantified via qPCR using the NEB Luna Universal qPCR Master Mix (NEB,M3003). We first assessed relative primer binding efficiencies through a series of 4 (1:10) serial dilutions of an aliquot of cDNA as in [47]. These efficiencies were then incorporated in the calculation of each oligo's relative expression (see below). For each transfection replicate we quantified the relative abundance of pDNA and cDNA for each oligo using a common forward primer binding within the GFP ORF and 12 different reverse primers, each targeting the oligo-specific barcode sequence. All reactions were performed in triplicate. Relative expression of each oligo for each transfection replicate was calculated with the  $\Delta C_t$  method, normalising it by the relative amount of pDNA. Expression differences between archaic and non-archaic alleles were then computed using the relative expression ratio, using a t-test to quantify any significant deviation from an expected ratio of 1 [48], BH-correcting the resulting p-values.

#### Results

#### Building a high-confidence set of archaic variants

To curate a list of high-confidence archaic SNPs to analyse for possible functional contributions to present- day Papuans, we took advantage of a recently described dataset of both archaic and non-archaic haplotypes segregating within 56 individuals of Papuan ancestry [6]. Due to incomplete lineage sorting and recombination, a substantial fraction of variants segregating in these archaic haplotypes is not expected to be of archaic origin [9]. Thus, to generate a high-confidence set of SNPs of archaic origin (aSNPs) we applied a series of filtering steps to our starting list of 7,791,042 genomic SNPs (see Methods and S1 Table). This allowed us to generate two high-confidence sets of 140,916 Denisovan and 88,625 Neanderthal aSNPs (S2 and S3 Tables, respectively). In parallel we defined a set of 229,541 (the sum of all high-confidence aSNPs) variants segregating in present-day Papuans outside archaic haplotypes and which likely arose after the Out of Africa event that matched the allele frequency distribution of our list of Denisovan and Neanderthal aSNPs (see Methods and S2(C) Fig) to act as a control set in all analyses. We refer to these variants as non-archaic SNPs (naSNPs, S4 Table).

Following calculation of the main introgressed allele frequency (MIAF) and of the derived allele frequency (DAF) for aSNPs and naSNPs respectively (see Methods), we found that 82,013 Denisova, 54,415 Neanderthal and 136,428 non-archaic variants are segregating at common-to-high-frequencies in PNG (MIAF or DAF > 0.05, see S1 Table). Importantly, we found that Denisovan and Neanderthal and non-archaic variants, whether segregating at low or common-to-high-frequencies, occur within genomic regions subjected to similar levels of background selection which are also higher (corresponding to less conserved regions) than those observed for protein-coding sequences (S3(A) and S3(B) Fig). We thus reasoned that, despite retaining only 10.5%, 9.2% and 4% of the original set of Denisovan, Neanderthal and non-archaic variants (S1 Table), our filtering criteria yielded a high-confidence set of archaic variants and an adequate corresponding background set of non-archaic alleles. Throughout the text and unless otherwise noted, to prioritise variants of potential evolutionary interest we focused on the above defined set of 82,013 Denisova, 54,415 Neanderthal and 136,428 non-archaic variants common-to-high-frequency variants.

### Similar patterns of introgression for Denisovan and Neanderthal aSNPs across chromatin states

To gain insights into the potential consequences of aSNPs in present-day Papuans, we began by annotating the set of aSNPs and naSNPs across multiple genomic elements (see Methods). We found that SNPs lie within non-coding elements, regardless whether considering the entire set of variants or only those instances segregating at MIAF or DAF  $\geq 0.05$  in New Guinea Island (S4(A) and S4(B) Fig). As previously reported [8, 9, 14], we also noted a significant depletion of both Denisovan and Neanderthal aSNPs from protein coding sequences, relatively to naSNPs (n = 3,431 and 1,919,  $p = 1.41 \cdot 10^{-11}$  and  $6.61 \cdot 10^{-19}$  for Denisovan and Neanderthal aSNPs, respectively).

To investigate the patterns of archaic introgression at a higher granularity, we intersected the set of common-to-high-frequency variants against epigenomic maps of 15 different chromatin states across 111 cell types generated by the Roadmap Epigenomics Project [34], and computed odds ratios (OR) between aSNPs and naSNPs to identify states containing an excess of Denisovan and/or Neanderthal variants. aSNPs are generally depleted from inactive chromatin states (states not associated with gene expression as defined by [34]) as well as promoters and transcription start sites, but enriched within transcribed and subsets of enhancer elements



Fig 1. Distribution of common-to-high-frequency aSNPs across 15 chromatin states and 111 cell types. Patterns of enrichment across the 15 chromatin states for the set of common-to-high-frequency Denisovan and Neanderthal aSNPs relatively to the matched background set of naSNPs. Histograms on top indicate the mean number of variants within each chromatin state across all 111 cell types. Asterisks indicate BH-corrected Fisher's exact test p-values: \* =  $p \le 0.05$ , \*\* =  $p \le 0.01$ ; \*\*\*\* =  $p \le 0.001$ ; \*\*\*\* =  $p \le 0.001$  (for full statistical results see S5 Table).

(Fig 1; full results are available at S5 Table). In particular, we observed a significant depletion of Denisovan and Neanderthal aSNPs from PolyComb repressed (ReprPC  $p_{Deni} = 7.7 \cdot 10^{-65}$ ), bivalent enhancers, promoters and transcription start sites (EnhBiv  $p_{Deni} = 5.0 \cdot 10^{-69}$ , BivFlnk  $p_{Deni} = 1.74 \cdot 10^{-20}$  and TssBiv  $p_{Deni} = 4.95 \cdot 10^{-20}$ , respectively). We also observed a Neander-thal-specific significant enrichment for heterochromatin (Het  $p_{Nean} = 2.9 \cdot 10^{-18}$ ), whereas Denisovan aSNPs are significantly depleted from this state ( $p_{Deni} = 4.6 \cdot 10^{-6}$ ). An exception to this trend was an excess of aSNPs from both archaic ancestries that fell within quiescent chromatin (Quies  $p_{Deni} = 1.5 \cdot 10^{-15}$ ), as previously reported by [12]. This observation is not surprising given that quiescent chromatin is the predominant state in each epigenome [34], and this finding might indicate that a substantial fraction of aSNPs are functionally inert in most contexts, although silencers have also been found enriched within this state [49].

When considering active chromatin states, we noticed a significant depletion of Denisovan and Neanderthal ancestries from transcription start sites (TssA  $p_{Deni} = 3.4 \cdot 10^{-4}$ ), promoters (TssAFlnk  $p_{Deni} = 0.026$ ) and states marked by both promoter and enhancer epigenetic marks (TxFlnk  $p_{Deni} = 0.0093$ ; Fig 1) and of Neanderthal aSNPs from enhancers (Enh,  $p_{Nean} = 1.3 \cdot 10^{-8}$ ). On the other hand, we observed a significant enrichment for Denisovan and Neander-thal aSNPs within genic enhancers (EnhG  $p_{Deni} = 1.6 \cdot 10^{-4}$ ) transcribed (Tx  $p_{Deni} = 3.9 \cdot 10^{-30}$ ) and within zinc-finger protein-genes and repetitive sequence (ZNF/Rpts  $p_{Deni} = 0.033$ )

states, this latter particularly for Neanderthal variants (Fig 1). Finally, we found a significant enrichment of Denisovan aSNPs within weakly transcribed regions (TxWk  $p_{Deni} = 2.3 \cdot 10^{-10}$ ) (Fig 1). Importantly, we did not find any substantial differences in these patterns when considering introgressed SNPs regardless of allele frequency (S5 Fig and S6 Table).

To validate this annotation approach we repeated our analyses using a set of 11,422,574 Neanderthal variants identified within 75 individuals of West Eurasian ancestry [31], which we processed as above. We found similar patterns of enrichment across chromatin states in the Papuan and West Eurasian datasets, with Neanderthal variants being significantly enriched within Quies, Het, Tx and TxWk states (S6(A) Fig). Similarly to [12], we also observed a significant depletion of Neanderthal variants from Enh and EnhBiv, although we detected a significant enrichment within EnhG (S6(A) Fig); however, the three states were collapsed into a single category in [12] and fewer SNPs were associated with this state than with Enh. These results not only validate our SNP filtering criteria; they also suggest that, on a genome-wide scale, different instances of admixture with archaic hominins might broadly contribute to modern human populations primarily through gene regulatory change.

Finally, to confirm whether these signals were caused by SNPs occurring within highly pleiotropic or cell-specific functional elements, we counted the number of different cell types across which aSNP and naSNP -containing elements were annotated with any given chromatin state. This yielded an estimate of the potential pleiotropic activity of each functional element. We observed similar degrees of pleiotropy across all the three ancestries, with variants within Quies, TssA, Tx and TxWk exhibiting higher pleiotropic activities than those within TssAFlnk, TxFlnk, Enh and EnhG (S7 Fig). Given that 64,944 Denisovan, 41,667 Neanderthal and 108,282 non-archaic variants fell within chromatin states associated with transcribed regions and CREs in at least one cell type (Fig 1 and S5 Fig), these results indicate that a non-trivial fraction of introgressed variants might have functional consequences, particularly for gene regulatory processes occurring in a limited number of cell types.

### Denisovan and Neanderthal aSNPs might have different effects on gene regulatory processes across tissues

Given the observed enrichment of aSNPs within chromatin states associated with CREs, we sought to investigate the effects of Denisovan and Neanderthal aSNPs on gene regulation. Hence, we restricted our analyses to all common-to-high-frequency SNPs within TssA, TssaFlnk, TxFlnk, Enh and EnhG chromatin states, which resulted in three sets of 39,636, 24,653 and 67,098 Denisovan, Neanderthal and non-archaic variants. Next, to identify tissues containing an over-representation of Denisovan and/or Neanderthal variants, for each tissue we again computed the OR between the number of aSNPs and naSNPs (see Methods). Despite roughly similar numbers of aSNPs falling within CREs across all tissue types, patterns of enrichment vary substantially by tissue type (Fig 2; for full results see S7 Table). For instance, Denisovan and Neanderthal aSNPs are significantly depleted from iPSCs ( $p_{Deni} = 1.2 \cdot 10^{-7}$ ,  $p_{Nean} = 9.6 \cdot 10^{-5}$ ) and Neurospheres ( $p_{Deni} = 0.02$ ,  $p_{Nean} = 0.0016$ ), but significantly enriched within mesenchymal ( $p_{Deni} = 1.7 \cdot 10^{-7}$ ,  $p_{Nean} = 3.3 \cdot 10^{-5}$ ) and blood & immune T cells ( $p_{Deni}$ =  $3.5 \cdot 10^{-15}$ ,  $p_{Nean} = 3.3 \cdot 10^{-6}$ ). In addition, enrichment patterns vary by aSNP ancestry, with Neanderthal-specific enrichment within smooth muscle ( $p_{Nean} = 8.2 \cdot 10^{-6}$ ) and Denisovan aSNPs significantly overrepresented within haematopoietic stem cells (HSCs) & B cells ( $p_{Deni}$  $= 5.0 \cdot 10^{-4}$ ).

As above, repeating this analysis in Neanderthal aSNPs segregating within Europeans also identified significant enrichments of Neanderthal aSNPs within CREs active in mesenchymal and blood & T cells (S6(B) Fig). Interestingly, these findings fully recapitulate previously



**Fig 2. Distribution of common-to-high-frequency aSNPs falling within CREs across 18 tissues.** Patterns of enrichment across the 18 different tissue types in Roadmap Epigenomics for the set of common-to-high-frequency aSNPs annotated within chromatin states associated with CREs, relatively to the matched background set of naSNPs. Histograms on top indicate the mean number of variants calculated across all cell types belonging to each tissue. Asterisks indicate BH-corrected Fisher's exact test p-values: \* =  $p \le 0.05$ , \*\* =  $p \le 0.001$ ; \*\*\*\* =  $p \le 0.001$ ; \*\*\*\* =  $p \le 0.001$  (for full statistical results see S7 Table).

reported observations in Europeans [11, 12]. Thus, in line with previous studies [11, 22, 50–52], our observation of a strong enrichment of Denisovan aSNPs within CREs active in immune cell types would suggest a potential contribution of these archaic variants to regulatory processes happening within immune-related cells, and by extension, to immune-related traits.

As a final test, we intersected these variants with significant cis-eQTLs (qval < 0.05) from GTEx v8 [35]. We found 59 Denisovan, 117 Neanderthal and 388 non-archaic variants (0.15%, 0.47% and 0.58% of all common-to-high-frequency SNPs, respectively) were identified as potential cis-eQTLs in at least one tissue in GTEx. We expanded the analysis to the entire set of variants, regardless of their allele frequency, and found 132 Denisovan, 242 Neanderthal and 688 non-archaic variants (0.09%, 0.27% and 0.3%, respectively) overlapping with cis-eQTLs. Given the well-recognised bias towards individuals of European ancestry within the GTEx dataset, these findings both support the need to include under-represented populations into genetic surveys, and reaffirm that the majority of our introgression assignments are correct, and, as expected, that our variants are geographically restricted.

#### Assessing the impact of aSNPs on transcription factor binding sites

We next sought to characterise whether aSNPs might alter gene regulation and the functional mechanisms by which this may occur. We thus examined the potential of aSNPs annotated

within CREs to disrupt known transcription factor binding sites (TFBS) by assessing their impact on 690 different position weight matrices (PWMs) drawn from the from HOCO-MOCO [40] and Jaspar [39] databases. After consolidating variants predicted to disrupt the same motif across the two databases (see Methods), we found that 16,048 Denisovan, 10,032 Neanderthal aSNPs and 28,370 non-archaic naSNPs (respectively 40.5%, 40.7% and 42.3% of the tested variants) were predicted to disrupt at least one TFBS. To avoid redundancy due to similarities between predicted motifs across closely related TFs, we took advantage of recent work by Vierstra *et al.* [41], which clusters HOCOCOMO and Jaspar PWMs into 286 distinct clusters on the basis of sequence similarity, and considered only those instances included in these clusters.

We then tested whether any of the motif clusters contained a significant excess of TFBS-disrupting aSNPs relative to naSNPs. Overall, we did not find any significant differences between the number of TFBS-disrupting archaic and non-archaic SNPs across the set of motif clusters (p = 0.26 and 0.14 for Denisovan and Neanderthal aSNPs, respectively). This suggests that, on a genome-wide scale, aSNPs are not disrupting any specific family of DNA motifs. To quantify the impact of aSNPs on the relevant PWM, for each of these TFBS-disrupting variants we then calculated a  $\Delta$  PWM score, defined as the difference in the PWM score between the archaic and non-archaic alleles in case of aSNPs or between the derived and ancestral allele in case of naSNPs. While naSNPs are on the whole predicted to be significantly more disruptive than aSNPs, this difference is likely caused by the number of TFBS-disrupting variants reported in each category as  $\Delta$  PWM scores remain highly comparable across ancestries (S8(B) Fig). Together, these results indicate that despite a lack of specific gene regulatory networks preferentially altered by aSNPs, a non-neglible fraction of our identified cis-regulatory variants might affect gene expression through altering the affinity between TFs and their cognate DNA sequences.

# High levels of population differentiation Denisovan aSNPs within CREs active in immune cells

Recent studies have shown the existence of genetic structure characterising the Indonesian archipelago, with Papuan—and thus, Denisovan—ancestry showing a marked west to east cline [6, 53]. We therefore investigated whether our subset of variants, particularly those within CREs active in immune-related cells, also showed similar levels of population differentiation across the region. To this extent, we defined aSNPs and naSNPs in haplotype data from 63 individuals from Western Indonesia, also part of [6]. From the total set of 14,117 Denisovan, 8,662 Neanderthal and 22,916 non-archaic variants annotated within CREs active in immune-related cells, we found that 4,732 Neanderthal and 9,682 non-archaic variants (42.2% and 54.6%, respectively) segregate within Neanderthal and non-archaic haplotypes in western Indonesias (Fig 3A). However, in keeping with the expectation of little to no Papuan genetic ancestry in Western Indonesia only 1,535 Denisovan variants (10.8%) in our Papuan sample are also segregating within Denisovan-like haplotypes in this additional data set (Fig 3A). Indeed, we found significantly higher  $F_{st}$  values for the set aSNPs annotated within CREs, especially those of Denisovan ancestry ( $p_{Wilcoxon} < 2.2 \cdot 10^{-16}$  and 0.0013, respectively for Denisovan variant (Neanderthal), compared to naSNPs (Fig 3B).

# Denisovan alleles are associated with gene expression differences in immune cells

Given the observed enrichment of aSNPs, especially of Denisovan ancestry, within the CREs of immune-related cells, we used GREAT [54] to link these to their nearest genes in an attempt





to identify possible targets. There was limited overlap across ancestries in the sets of putative target genes. For instance, out of a total number of 9,504 putatively target genes only 303 (3.2%) are predicted to be targeted by all three ancestries (Fig 4A). Conversely, we found that 795 (8.3%) and 573 (6.0%) genes are uniquely targeted by Denisovan and Neanderthal variants, respectively (Fig 4A).

We next performed a Gene Ontology (GO) enrichment analysis on the set of genes associated with either archaic ancestry, using the set of genes associated with naSNPs as a bacground set. Denisovan aSNPs are associated with genes strongly involved in multiple immune-related processes (S8 Table). Genes targeted by Neanderthal variants are instead enriched for more general biological processes, albeit instances related to neutrophil/granulocyte migration and chemotaxis were observed for Neanderthal variants (S8 Table). We then computed the semantic similarity between the significantly enriched terms (GREAT reported BH-adjusted hypergeometric  $p \le 0.01$ ) using a method that incorporates both the locations of the terms in the GO graph as well as their relations with their ancestor terms [55], and found relatively low levels of semantic similarity between archaic and non-archaic enriched GO terms (Fig 4B), suggesting the set of variants might indeed affect the regulation of different biological processes within immune-related cells. Finally, for each ancestry we manually examined the set of genes associated with the significant GO terms. Denisovan cis-regulatory variants were predicted to regulate genes such as TNFAIP3, OAS2 and OAS3, all of which have been repeatedly identified as harbouring archaic hominin contribution that impact immune responses to pathogens [51, 52, 56]. In particular, we found 20 Denisovan variants associated with OAS2 and OAS3, eight of which (rs368816473, rs372433785, rs139804868, rs146859513, rs143462183, rs370655920, rs375463218 and rs372139279) were also predicted to significantly alter the affinity between TFs and their cognate DNA sequences by our analyses. Notably, in all cases the archaic allele segregated at frequencies between 0.2 and 0.4 in Papuans but was absent from Western





Indonesia. Similarly, a comparison with 6 different non-human primate genomes indicated that for 7 of these 8 Denisovan aSNPs the introgressed allele is derived.

The OAS locus has been repeatedly found to harbour signals of both Neanderthal and Denisovan adaptive introgression, with archaic alleles predicted to alter gene regulation [52, 56]. Our previous work has shown that both OAS2 and OAS3 are differentially expressed in whole blood between the people of Mentawai, a small barrier island off the coast of Sumatra, in West Indonesia, and the Korowai, a genetically Papuan group living on the Indonesian side of New Guinea island (Fig 5B) [57]. To understand whether Denisovan introgression might contribute to these differences, we tested the regulatory activity of five of the eight Denisovan variants described above (rs372433785, rs139804868, rs146859513, rs143462183 and rs370655920), all of which were predicted to disrupt the binding sites of TFs active in human immune cells, using a plasmid reporter experiment (see Methods).

We tested all five alleles across two different lymphoblastoid cell lines (LCLs) established from Papuan donors. In all cases, we found the direction of effect to be consistent across biological and technical replicates. In particular, two Denisovan alleles (rs139804868:A>G and rs146859513:C>G) consistently showed significantly lower transcriptional rates compared to their non-archaic counterpart. rs139804868:A>G is predicted to disrupt a motif bound by *BHLHA15* and *TAL1*, which have been respectively found to be involved in immune B cell differentiation [58] and hematopoiesis [59]. rs146859513:C>G is predicted to disrupt a binding site for *NFKB2*, which is known to regulate the expression of cytokines and chemokines related genes [60] (Fig 5C). While confirming the validity of our approach, taken together, these findings point to a substantial contribution of Denisovan variants to immune-related processes in present-day Papuans, one chiefly mediated through the regulation of active immune responses mounted against pathogenic infections.



Fig 5. Functional validation of the regulatory impact of Denisovan variants near OAS2 and OAS3. A) The genomic region encompassing the eight Denisovan variants associated with OAS2 and OAS3. The top two tracks display patterns of DNase Hypersensitivity sites in Blood and immune T cells as well as in HSC and B cells [34]. Tested variants are shown along with their calculated  $\Delta$  PWM. Bottom tracks display the chromatin state information for the same tissues; **B**) the distribution of the log<sub>2</sub> RNA-seq counts per million in whole blood for OAS2 and OAS3 between the Korowai and the people of Mentawai, from [57]; **C**) the relative expression changes between Denisovan and non-archaic alleles, or between the alternative and the reference allele for positive control (rs9283753) [45], in two Papuan LCLs. Asterisks mark significant differences from 1, BH-corrected p < 0.05: \*\*\* = 0.0001 <  $p \le 0.001$ ; \*\* = 0.001 <  $p \le 0.001$ ; \*\* = 0.001 <  $p \le 0.05$ .

#### Discussion

There is significant interest in understanding the functional consequences of archaic introgression. Evidence indicates that both Denisovan and Neanderthal aSNPs, especially those within protein coding and conserved non-coding elements, are mostly deleterious and negatively selected in modern humans [14, 61]. Similar findings have been recently reported for highly pleiotropic enhancers, where aSNPs are depleted likely as a consequence of their potential to perturb gene expression across multiple tissues [12]. Nevertheless, out of the substantial number of archaic variants still segregating within present-day populations, a large fraction falls within genomic regions that show strong evidence of functional activity across a variety of cell types. Indeed, studies conducted primarily on Neanderthal introgressed DNA have suggested a non-negligible contribution to gene expression variation in modern humans [10, 11, 22], with repeated examples of Neanderthal archaic variants falling within regulatory elements or the seed region of mature micro-RNAs predicted to affect transcriptional and post-transcriptional regulatory processes [11, 62].

In this study, we have taken advantage of a recently published dataset [6] to investigate the landscape of archaic introgression in individuals of Papuan genetic ancestry, the functional consequences of which remained poorly understood. This has allowed us to characterise the putative contribution of Denisovan DNA, which is known to account for up to 5% of the genome of present-day Papuans [4, 63], while also comparing it to that of Neanderthal DNA and non-archaic variants that arose following the Out of Africa event. We specifically analysed all of our variants across multiple cell types and functional chromatin states aiming to account for the strong dependency on the cells' chromatin landscapes [34] when assessing the potential activities of introgressed alleles. While our analyses focus on a small subset of introgressed variants identified by [6], our filtering criteria ensure that we only consider variants with a high likelihood of being truly introgressed. In addition, comparing Denisovan and Neanderthal signals within the same samples serves as an internal control for difference in linkage disequilibrium and background selection between ancestries, hence providing stronger support for any eventual Denisovan- and/or Neanderthal-specific contribution to phenotypic variation in contemporary Papuans.

First, we confirm previous reports that aSNPs mostly occur within non-coding sequences [9, 14, 64], with a large number falling within highly constitutive and/or functionally inert regions, which might be the result of weaker purifying selection acting on these elements. aSNPs are significantly depleted from other inactive chromatin states but over-represented within states putatively containing elements involved in gene regulatory processes. Applying our approach to Neanderthal aSNPs segregating within present-day West Eurasians [31] replicates most of our findings, as well as previously reported signals including a depletion of Neanderthal variants within enhancer-like elements [12]. Previous studies have shown that, while polygenic risk scores [65] and eQTLs [66] have limited cross-population portability, biochemically active functional regulatory elements tend to be under evolutionary constraint within the human lineage [67, 68]. Thus, while our approach is contingent on functional data generated in a population that is not of Papuan ancestry, our focus on functional analyses are likely to be more robust to differences across populations in linkage disequilibrium. Nevertheless, our set of SNPs, and our conclusions, represent a first step that will benefit from functional validation at scale.

Second, we find notable differences between the two archaic ancestries in their patterns of introgression. Such differences are likely consequence of the tissue-specificity nature of CREs, and of enhancers in particular [69]. Indeed we report that variants annotated in these CRE-associated states tend to lie within elements active in a restricted number of cell types, suggesting that the effects of archaic introgression might be highly cell-type-specific. In addition, admixture with Neanderthals might have similar consequences even across distantly related human populations, with Neanderthal aSNPs strongly enriched within elements active within blood & immune T cells and mesenchymal cells both in Papuans and Europeans, replicating previously reported observations [11, 12]. While we report a similar enrichment for Denisovan alleles within these tissues, we further note a Denisovan aSNPs-specific significant enrichment with CREs active within HSC & B cells, suggesting a more pervasive contribution of Denisovan introgression to gene regulatory processes happening within immune-related cells. In line

with this, we further show that Denisovan and Neanderthal variants annotated within immune-related CREs might target different sets of genes with limited overlap in the biological processes they are involved in. Indeed, only genes predicted to be regulated by Denisovan aSNPs are strongly involved in active immune responses.

Third, we have characterised the impact of introgressed variants on transcription factor binding sites. A substantial proportion of aSNPs within CREs are predicted to modify the interactions between TFs and their cognate binding sites, although we do not detect an overall higher impact of aSNPs relative to naSNPs. Our data suggest that, despite aSNPs possibly disrupting individual DNA motifs, at a genome-wide level admixture with archaic hominins is unlikely to have resulted in large rewiring of transcription factor regulatory networks. While this might contrast the findings of previous studies [70], in our study we focused on families of DNA motifs (as clustered by [41]) rather than consider enrichment within individual PWMs, an approach that we believe is better suited to controlling for the typical redundancy of PWMs. This, however does not exclude that aSNPs might not disrupt specific DNA motifs. Indeed, our approach also identifies Neanderthal aSNPs disrupting the PWM of TFs such as *CREB1, USF1, MYB* and *STAT6*, similarly to what has been reported by [70]. We also note that there is a significant (hypergeometric  $p < 2.2 \cdot 10^{-16}$ ) overlap between genes targeted by N-eQTLs in [70] and our own set of possible target genes (S9 Fig).

Finally, we performed a plasmid reporter assay experiment to quantify the molecular impact of Denisovan TFBS-disrupting variants annotated within CREs active in immunerelated cells and predicted to affect the expression level of OAS2 and OAS3. These genes belong to a family of pattern-recognition receptors involved in innate immune responses against both RNA and DNA viruses, with OAS3 considered to be essential in reducing viral titer during Chikungunya, Sindbis, influenza or vaccinia viral infections [71]. At least two previous studies have shown the presence of both Neanderthal and Denisovan archaic haplotypes segregating at this locus respectively within European [52] and Papuan [56] individuals. Sams et al. found two variants (rs10774671, rs1557866) within these Neanderthal haplotypes which are respectively associated with the codification of different OAS1 splicing isoforms and with a reduction in OAS3 expression levels, the latter only upon viral infection [52]. Our previous work has found that both OAS2 and OAS3 are differentially expressed between western Indonesians and Papuans [57]. Here we report a set of eight SNPs (rs368816473, rs372433785, rs139804868, rs146859513, rs143462183, rs370655920, rs375463218 and rs372139279), located roughly 41 kb upstream of OAS2 and OAS3, all of which are predicted to strongly alter the ability of different transcription factors, including IRF4, NFKB2 and TAL1, to bind to their underlying DNA motifs. In all cases, the reference allele is fixed within western Indonesian populations, whereas the archaic alleles segregate at frequencies between 0.2 and 0.4 in Papuans. We show that at least 5 of these variants lie within sequences that can regulate expression in reporter gene plasmid assays, and that in two of these variants (rs139804868 and rs146859513) the Denisovan allele is associated with significantly lower transcriptional activity compared to the non-archaic allele in immune cells from two different Papuan donors, again suggesting that these SNPs are of biological importance.

Recent work by [72] has shown that variants which arose in the human lineage following its split from the Neanderthal-Denisovan common ancestor, and which have reached (near) fixation in modern humans since then can exhibit significant differences in regulatory activity relative to the ancestral state. Similarly, Jagoda *et al.* has shown large impact on gene expression associated with the presence of Neanderthal aSNPs in vitro [73]. Our results suggest that Denisovan alleles segregating within modern human populations are also likely to actively participate in gene regulatory processes, especially those that take place within immune-related cells. This agrees with recent findings from a study that analysed the genome of present-day people

of Taiwan, the Philippines, the Solomon Islands and Vanuatu [50]. While further experimental validation of our observations is necessary in order to characterise the genome-wide impact of archaic introgression, the results presented in this study argue for a potential contribution of Denisovan variants to immune-related phenotypes amongst early modern humans in the region, potentially favouring adaptation to the local environment [15, 22].

#### Supporting information

**S1** Table. Table reporting the number of SNPs retained after each filtering step. (XLSX)

**S2** Table. File reporting the genomic location of all Denisovan variants used in this study. (TXT)

S3 Table. File reporting the genomic location of all Neanderthal variants used in this study.

(TXT)

**S4** Table. File reporting the genomic location of all non-archaic variants used in this study. (TXT)

S5 Table. Table reporting the enrichment results for the set of common-to-high-frequency archaic variants annotated within each chromatin state. (XLSX)

S6 Table. Table reporting the enrichment results for the entire set of archaic variants annotated within each chromatin state.

(XLSX)

S7 Table. Table reporting the enrichment results across tissues for the set of common-tohigh-frequency archaic variants annotated within chromatin states associated to CREs. (XLSX)

S8 Table. Table reporting the GO enrichment results for the set of genes targeted by Denisovan, Neanderthal and non-archaic variants. (XLSX)

**S9** Table. Table reporting the primer and oligo sequences used in the *in-vitro* validation. (XLSX)

**S1 Fig. aSNPs distribution across haplotypes.** Histogram showing the distribution of the Denisovan and Neanderthal allele frequency differences between archaic and non-archaic haplotypes. Negative values indicate putative archaic variants segregating at higher frequencies within non-archaic haplotypes. (PDF)

**S2 Fig. Overview of site frequency spectrum (SFS).** Histogram showing the SFS for Denisovan, Neanderthal and non-archaic variants **A**) before and **B**) after the variant filtering steps. **C**) Histogram showing the SFS for aSNPs (Neanderthal + Denisovan) and the matched background set of naSNPs.

(PDF)

**S3 Fig. Estimates of the levels of background selection.** Plots showing the distribution of the B-statistic values for the genomic regions containing the refined set of **A**) all Denisovan, Nean-derthal and non-archaic variants or **B**) only the common-to-high-frequency SNPs. Lower

values indicate higher evolutionary constrains. For visual comparison, the distribution of the B-statistic values for all human protein coding sequences is also shown. Reported p-values are calculated from Mann-Whitney U test.

(PDF)

**S4 Fig. Genome-wide functional annotation of aSNPs and naSNPs.** Histogram showing the proportion of **A**) all aSNPs and naSNPs and **B**) only common-to-high-frequency variants annotated across multiple genomic elements as reported by annotatr [64]. Percentages are relative to the SNPs annotated across the categories shown. Total numbers of variant annotated: **A**) 72,761 Denisovan, 44,135 Neanderthal and 114,518 non-archaic; **B**) 42,334 Denisovan, 27,400 Neanderthal and 68,310 non-archaic. (PDF)

**S5 Fig. Distribution of all aSNPs across 15 chromatin states.** Figure showing the patterns of enrichment across the 15 chromatin states for the entire set of Denisovan and Neanderthal aSNPs relatively to the matched background set of naSNPs. Histograms on top indicate the mean number of variants within each chromatin state calculated across all 111 cell types. Asterisks indicate BH-corrected Fisher's exact test p-values < 0.05, i.e., \*\*\*\* =  $p \le 0.0001$ ; \*\*\* = 0.0001 ; \*\*\* = 0.001 <math>; \*\* = 0.001 <math>; \*\* = 0.001 <math>; \*\*\* = 0.001 <math>



S6 Fig. Distribution of Neanderthal aSNPs identified within EUR across chromatin states and tissues. Figure showing the patterns of enrichment across A) the 15 chromatin states for the entire set of Neanderthal aSNPs B) the 18 different tissues for the set of cis-regulatory Neanderthal variants segregating at common-to-high-frequencies in EUR. OR are computed relatively to the matched background set of naSNPs. Histograms on top indicate the mean number of variants annotated within each chromatin state (A) or within each tissue (B). The mean is respectively calculated across all 111 cell types and across the cell types belonging to each tissue. Asterisks indicate BH-corrected Fisher's exact test p-values < 0.05, i.e., \*\*\*\* =  $p \le 0.0001$ ; \*\*\* = 0.0001 <  $p \le 0.001$ ; \*\* = 0.001 <  $p \le 0.01$ ; \*\* = 0.01 <  $p \le 0.05$ . (PDF)

**S7 Fig. Pleiotropic activities of chromatin state-associated elements carrying SNPs.** Cumulative proportion of the pleiotropic activity across 111 cell types of each chromatin state-associated element carrying SNPs. (PDF)

**S8 Fig. SNPs impact on TFBSs.** Distribution of the  $\Delta$  PWM scores for the set of aSNPs and naSNPs. P-values are returned from Wilcoxon test. (PDF)

**S9** Fig. Overlap between target genes identified by Findley et al and this study. (PDF)

#### Acknowledgments

We would like to thank all members of the Gallego Romero and McCarthy groups for their helpful discussions and comments on the manuscript. We also wish to acknowledge all of the study participants who generously consented to genome sequencing in the original study, and the leadership of Herawati Sudoyo and Chelzie Crenna Darusalam (Eijkman Institute for Molecular Biology) in generating these datasets.

#### **Author Contributions**

Conceptualization: Davide M. Vespasiani, Irene Gallego Romero.

Data curation: Guy S. Jacobs.

Formal analysis: Davide M. Vespasiani, Guy S. Jacobs.

Funding acquisition: François-Xavier Ricaut, Irene Gallego Romero.

Investigation: Davide M. Vespasiani, Laura E. Cook, Nicolas Brucato.

Methodology: Davide M. Vespasiani, Laura E. Cook, Nicolas Brucato.

Project administration: Nicolas Brucato, Irene Gallego Romero.

**Resources:** Guy S. Jacobs, Nicolas Brucato, Matthew Leavesley, Christopher Kinipi, François-Xavier Ricaut, Murray P. Cox.

Supervision: Murray P. Cox, Irene Gallego Romero.

Writing - original draft: Davide M. Vespasiani, Irene Gallego Romero.

Writing – review & editing: Davide M. Vespasiani, Guy S. Jacobs, Laura E. Cook, Nicolas Brucato, Matthew Leavesley, Christopher Kinipi, François-Xavier Ricaut, Murray P. Cox, Irene Gallego Romero.

#### References

- Green RE, Krause J, Briggs AW, Maricic T, Stenzel U, Kircher M, et al. A Draft Sequence of the Neandertal Genome. Science. 2010; 328(5979):710–723. https://doi.org/10.1126/science.1188021 PMID: 20448178
- Reich D, Green RE, Kircher M, Krause J, Patterson N, Durand EY, et al. Genetic history of an archaic hominin group from Denisova cave in Siberia. Nature. 2010; 468(7327):1053–1060. <u>https://doi.org/10. 1038/nature09710 PMID: 21179161</u>
- Lorente-Galdos B, Lao O, Serra-Vidal G, Santpere G, Kuderna LFK, Arauna LR, et al. Whole-genome sequence analysis of a Pan African set of samples reveals archaic gene flow from an extinct basal population of modern humans into sub-Saharan populations. Genome Biology. 2019; 20(1):1–15. https:// doi.org/10.1186/s13059-019-1684-5 PMID: 31023378
- Reich D, Patterson N, Kircher M, Delfin F, Nandineni MR, Pugach I, et al. Denisova Admixture and the First Modern Human Dispersals into Southeast Asia and Oceania. American Journal of Human Genetics. 2011; 89:516–528. https://doi.org/10.1016/j.ajhg.2011.09.005 PMID: 21944045
- Larena M, McKenna J, Sanchez-Quinto F, Bernhardsson C, Ebeo C, Reyes R, et al. Philippine Ayta possess the highest level of Denisovan ancestry in the world. Current Biology. 2021; 31(19):4219– 4230.e10. https://doi.org/10.1016/j.cub.2021.07.022 PMID: 34388371
- Jacobs GS, Hudjashov G, Saag L, Kusuma P, Darusallam CC, Lawson DJ, et al. Multiple Deeply Divergent Denisovan Ancestries in Papuans. Cell. 2019; p. 1010–1021. https://doi.org/10.1016/j.cell.2019. 02.035 PMID: 30981557
- Browning SR, Browning BL, Zhou Y, Tucci S, Akey JM. Analysis of Human Sequence Data Reveals Two Pulses of Archaic Denisovan Admixture. Cell. 2018; 173(1):53–61.e9. https://doi.org/10.1016/j. cell.2018.02.031 PMID: 29551270
- Sankararaman S, Mallick S, Dannemann M, Prüfer K, Kelso J, Pääbo S, et al. The genomic landscape of Neanderthal ancestry in present-day humans. Nature. 2014; 507(7492):354–357. <u>https://doi.org/10. 1038/nature12961</u> PMID: 24476815
- Dannemann M, Prüfer K, Kelso J. Functional implications of Neandertal introgression in modern humans. Genome Biology. 2017; 18(1):1–11. <u>https://doi.org/10.1186/s13059-017-1181-7</u> PMID: 28366169
- McCoy RC, Wakefield J, Akey JM. Impacts of Neanderthal-Introgressed Sequences on the Landscape of Human Gene Expression. Cell. 2017; 168(5):916–927.e12. <u>https://doi.org/10.1016/j.cell.2017.01.</u> 038 PMID: 28235201

- Silvert M, Quintana-Murci L, Rotival M. Impact and Evolutionary Determinants of Neanderthal Introgression on Transcriptional and Post-Transcriptional Regulation. The American Journal of Human Genetics. 2019; 104(6):1241–1250. https://doi.org/10.1016/j.ajhg.2019.04.016 PMID: 31155285
- Telis N, Aguilar R, Harris K. Selection against archaic hominin genetic variation in regulatory regions. Nature Ecology & Evolution. 2020; 4(11):1558–1566. <u>https://doi.org/10.1038/s41559-020-01284-0</u> PMID: 32839541
- Sankararaman S, Mallick S, Patterson N, Reich D. The Combined Landscape of Denisovan and Neanderthal Ancestry in Present-Day Humans. Current Biology. 2016; 26(9):1241–1247. https://doi.org/10. 1016/j.cub.2016.03.037 PMID: 27032491
- Petr M, Pääbo S, Kelso J, Vernot B. Limits of long-term selection against Neandertal introgression. Proceedings of the National Academy of Sciences of the United States of America. 2019; 116(5):1639–1644. https://doi.org/10.1073/pnas.1814338116 PMID: 30647110
- Dannemann M, Kelso J. The Contribution of Neanderthals to Phenotypic Variation in Modern Humans. American Journal of Human Genetics. 2017; 101(4):578–589. https://doi.org/10.1016/j.ajhg.2017.09. 010 PMID: 28985494
- McArthur E, Rinker DC, Capra JA. Quantifying the contribution of Neanderthal introgression to the heritability of complex traits. Nature Communications. 2021; 12(1):4481. https://doi.org/10.1038/s41467-021-24582-y PMID: 34294692
- Gittelman RM, Schraiber JG, Vernot B, Mikacenic C, Wurfel MM, Akey JM. Archaic Hominin Admixture Facilitated Adaptation to Out-of-Africa Environments. Current Biology. 2016; 26(24):3375–3382. https://doi.org/10.1016/j.cub.2016.10.041 PMID: 27839976
- Dolgova O, Lao O. Evolutionary and medical consequences of archaic introgression into modern human genomes. Genes. 2018; 9(7). https://doi.org/10.3390/genes9070358 PMID: 30022013
- Gouy A, Excoffier L. Polygenic Patterns of Adaptive Introgression in Modern Humans Are Mainly Shaped by Response to Pathogens. Molecular Biology and Evolution. 2020; 37(5):1420–1433. <a href="https://doi.org/10.1093/molbev/msz306">https://doi.org/10.1093/molbev/msz306</a> PMID: 31935281
- Enard D, Petrov DA. Evidence that RNA Viruses Drove Adaptive Introgression between Neanderthals and Modern Humans. Cell. 2018; 175(2):360–371.e13. https://doi.org/10.1016/j.cell.2018.08.034 PMID: 30290142
- Deschamps M, Laval G, Fagny M, Itan Y, Abel L, Casanova JL, et al. Genomic Signatures of Selective Pressures and Introgression from Archaic Hominins at Human Innate Immunity Genes. American Journal of Human Genetics. 2016; 98(1):5–21. https://doi.org/10.1016/j.ajhg.2015.11.014 PMID: 26748513
- Quach H, Rotival M, Pothlichet J, Loh YHE, Dannemann M, Zidane N, et al. Genetic Adaptation and Neandertal Admixture Shaped the Immune System of Human Populations. Cell. 2016; 167(3):643–656. e17. https://doi.org/10.1016/j.cell.2016.09.024 PMID: 27768888
- Hindorff LA, Bonham VL, Brody LC, Ginoza MEC, Hutter CM, Manolio TA, et al. Prioritizing diversity in human genomics research. Nature Reviews Genetics. 2018; 19(3):175–185. <u>https://doi.org/10.1038/</u> nrg.2017.89 PMID: 29151588
- Popejoy AB, Fullerton SM. Genomics is failing on diversity. Nature. 2016; 538(7624):161–164. <u>https://doi.org/10.1038/538161a PMID: 27734877</u>
- Simonti CN, Vernot B, Bastarache L, Bottinger E, Carrell DS, Chisholm RL, et al. The phenotypic legacy of admixture between modern humans and Neandertals. Science. 2016; 351(6274):737–741. <u>https://</u> doi.org/10.1126/science.aad2149 PMID: 26912863
- Franchini LF, Pollard KS. Human evolution: the non-coding revolution. BMC Biology. 2017; 15(1):89. https://doi.org/10.1186/s12915-017-0428-9 PMID: 28969617
- Gallego Romero I, Ruvinsky I, Gilad Y. Comparative studies of gene expression and the evolution of gene regulation. Nature Reviews Genetics. 2012; 13(7):505–516. https://doi.org/10.1038/nrg3229
- Herrero J, Muffato M, Beal K, Fitzgerald S, Gordon L, Pignatelli M, et al. Ensembl comparative genomics resources. Database. 2016; 2016:bav096. <u>https://doi.org/10.1093/database/bav096</u> PMID: 26896847
- Prüfer K, Racimo F, Patterson N, Jay F, Sankararaman S, Sawyer S, et al. The complete genome sequence of a Neanderthal from the Altai Mountains. Nature. 2014; 505(7481):43–49. https://doi.org/ 10.1038/nature12886 PMID: 24352235
- **30.** The 1000 Genomes Project Consortium. A global reference for human genetic variation. Nature. 2015; 526(7571):68–74. https://doi.org/10.1038/nature15393
- Mallick S, Li H, Lipson M, Mathieson I, Gymrek M, Racimo F, et al. The Simons Genome Diversity Project: 300 genomes from 142 diverse populations. Nature. 2016; 538(7624):201–206. <u>https://doi.org/10.1038/nature18964</u> PMID: 27654912

- McVicker G, Gordon D, Davis C, Green P. Widespread Genomic Signatures of Natural Selection in Hominid Evolution. PLOS Genetics. 2009; 5(5):e1000471. https://doi.org/10.1371/journal.pgen. 1000471 PMID: 19424416
- Cavalcante RG, Sartor MA. Annotatr: Genomic regions in context. Bioinformatics. 2017; 33(15):2381– 2383. https://doi.org/10.1093/bioinformatics/btx183 PMID: 28369316
- Roadmap Epigenomics Consortium. Integrative analysis of 111 reference human epigenomes. Nature. 2015; 518(7539):317–329. https://doi.org/10.1038/nature14248
- Consortium TG. The GTEx Consortium atlas of genetic regulatory effects across human tissues. Science. 2020; 369:1318–1330. https://doi.org/10.1126/science.aaz1776
- Lawrence M, Gentleman R, Carey V. rtracklayer: an R package for interfacing with genome browsers. Bioinformatics. 2009; 25(14):1841–1842. <u>https://doi.org/10.1093/bioinformatics/btp328</u> PMID: 19468054
- Benjamini Y, Hochberg Y. Controlling the False Discovery Rate: A Practical and Powerful Approach to Multiple Testing. Journal of the Royal Statistical Society Series B (Methodological). 1995; 57(1):289– 300. https://doi.org/10.1111/j.2517-6161.1995.tb02031.x
- Coetzee SG, Coetzee GA, Hazelett DJ. motifbreakR: an R/Bioconductor package for predicting variant effects at transcription factor binding sites. Bioinformatics. 2015; 31(23):3847–3849. <u>https://doi.org/10.1093/bioinformatics/btv470 PMID: 26272984</u>
- Khan A, Fornes O, Stigliani A, Gheorghe M, Castro-Mondragon JA, Van Der Lee R, et al. JASPAR 2018: Update of the open-access database of transcription factor binding profiles and its web framework. Nucleic Acids Research. 2018; 46(D1):D260–D266. <u>https://doi.org/10.1093/nar/gkx1126</u> PMID: 29140473
- 40. Kulakovskiy IV, Vorontsov IE, Yevshin IS, Sharipov RN, Fedorova AD, Rumynskiy EI, et al. HOCO-MOCO: towards a complete collection of transcription factor binding models for human and mouse via large-scale ChIP-Seq analysis. Nucleic Acids Research. 2018; 46(D1):D252–D259. <u>https://doi.org/10. 1093/nar/gkx1106</u> PMID: 29140464
- Vierstra J, Lazar J, Sandstrom R, Halow J, Lee K, Bates D, et al. Global reference mapping of human transcription factor footprints. Nature. 2020; 583(7818):729–736. https://doi.org/10.1038/s41586-020-2528-x PMID: 32728250
- 42. Shannon P, Richards M. MotifDb: An Annotated Collection of Protein-DNA Binding Sequence Motifs; 2020.
- 43. Gu Z. rGREAT: Client for GREAT Analysis; 2020.
- Ashburner M, Ball CA, Blake JA, Botstein D, Butler H, Cherry JM, et al. Gene Ontology: tool for the unification of biology. Nature Genetics. 2000; 25(1):25–29. https://doi.org/10.1038/75556 PMID: 10802651
- 45. Tewhey R, Kotliar D, Park DS, Liu B, Winnicki S, Reilly SK, et al. Direct Identification of Hundreds of Expression-Modulating Variants using a Multiplexed Reporter Assay. Cell. 2016; 165(6):1519–1529. https://doi.org/10.1016/j.cell.2016.04.027 PMID: 27259153
- 46. Klein JC, Agarwal V, Inoue F, Keith A, Martin B, Kircher M, et al. A systematic evaluation of the design and context dependencies of massively parallel reporter assays. Nature Methods. 2020; 17(11):1083– 1091. https://doi.org/10.1038/s41592-020-0965-y PMID: 33046894
- Svec D, Tichopad A, Novosadova V, Pfaffl MW, Kubista M. How good is a PCR efficiency estimate: Recommendations for precise and robust qPCR efficiency assessments. Biomolecular Detection and Quantification. 2015; 3:9–16. https://doi.org/10.1016/j.bdq.2015.01.005 PMID: 27077029
- **48.** Pfaffl MW. A new mathematical model for relative quantification in real-time RT-PCR. Nucleic Acids Research. 2001; 29(9):45e–45. https://doi.org/10.1093/nar/29.9.e45 PMID: 11328886
- Doni Jayavelu N, Jajodia A, Mishra A, Hawkins RD. Candidate silencer elements for the human and mouse genomes. Nature Communications. 2020; 11(1):1061. <u>https://doi.org/10.1038/s41467-020-14853-5 PMID: 32103011</u>
- Choin J, Mendoza-Revilla J, Arauna LR, Cuadros-Espinoza S, Cassar O, Larena M, et al. Genomic insights into population history and biological adaptation in Oceania. Nature. 2021; 592(7855):583–589. https://doi.org/10.1038/s41586-021-03236-5 PMID: 33854233
- Zammit NW, Siggs OM, Gray PE, Horikawa K, Langley DB, Walters SN, et al. Denisovan, modern human and mouse TNFAIP3 alleles tune A20 phosphorylation and immunity. Nature Immunology. 2019; 20(10):1299–1310. https://doi.org/10.1038/s41590-019-0492-0 PMID: 31534238
- Sams AJ, Dumaine A, Nédélec Y, Yotova V, Alfieri C, Tanner JE, et al. Adaptively introgressed Neandertal haplotype at the OAS locus functionally impacts innate immune responses in humans. Genome Biology. 2016; 17(1):1–15. https://doi.org/10.1186/s13059-016-1098-6 PMID: 27899133

- Hudjashov G, Karafet TM, Lawson DJ, Downey S, Savina O, Sudoyo H, et al. Complex Patterns of Admixture across the Indonesian Archipelago. Molecular Biology and Evolution. 2017; 34(10):2439– 2452. https://doi.org/10.1093/molbev/msx196 PMID: 28957506
- McLean CY, Bristor D, Hiller M, Clarke SL, Schaar BT, Lowe CB, et al. GREAT improves functional interpretation of cis-regulatory regions. Nature biotechnology. 2010; 28(5):495–501. <u>https://doi.org/10.1038/nbt.1630</u> PMID: 20436461
- 55. Wang JZ, Du Z, Payattakool R, Yu PS, Chen CF. A new method to measure the semantic similarity of GO terms. Bioinformatics. 2007; 23(10):1274–1281. https://doi.org/10.1093/bioinformatics/btm087 PMID: 17344234
- Mendez FL, Watkins JC, Hammer MF. Global genetic variation at OAS1 provides evidence of archaic admixture in Melanesian populations. Molecular Biology and Evolution. 2012; 29(6):1513–1520. <u>https:// doi.org/10.1093/molbev/msr301</u> PMID: 22319157
- 57. Natri HM, Bobowik KS, Kusuma P, Darusallam CC, Jacobs GS, Hudjashov G, et al. Genome-wide DNA methylation and gene expression patterns reflect genetic ancestry and environmental differences across the Indonesian archipelago. PLoS Genetics. 2020; 16(5):1–21. https://doi.org/10.1371/journal.pgen.1008749 PMID: 32453742
- Capoccia BJ, Lennerz JKM, Bredemeyer AJ, Klco JM, Frater JL, Mills JC. Transcription factor MIST1 in terminal differentiation of mouse and human plasma cells. Physiol Genomics. 2011; 43(3):174–186. https://doi.org/10.1152/physiolgenomics.00084.2010 PMID: 21098683
- Porcher C, Chagraoui H, Kristiansen MS. SCL/TAL1: a multifaceted regulator from blood development to disease. Blood. 2017; 129(15):2051–2060. https://doi.org/10.1182/blood-2016-12-754051 PMID: 28179281
- Li Q, Verma IM. NF-κB regulation in the immune system. Nature Reviews Immunology. 2002; 2 (10):725–734. https://doi.org/10.1038/nri910 PMID: 12360211
- Harris K. Evidence for recent, population-specific evolution of the human mutation rate. Proceedings of the National Academy of Sciences. 2015; 112(11):3439–3444. https://doi.org/10.1073/pnas. 1418652112 PMID: 25733855
- Lopez-Valenzuela M, Ramirez O, Rosas A, Garcia-Vargas S, de la Rasilla M, Lalueza-Fox C, et al. An Ancestral miR-1304 Allele Present in Neanderthals Regulates Genes Involved in Enamel Formation and Could Explain Dental Differences with Modern Humans. Molecular Biology and Evolution. 2012; 29 (7):1797–1806. https://doi.org/10.1093/molbev/mss023 PMID: 22319171
- Vernot B, Tucci S, Kelso J, Schraiber JG, Wolf AB, Gittelman RM, et al. Excavating Neandertal and Denisovan DNA from the genomes of Melanesian individuals. Science. 2016;. <u>https://doi.org/10.1126/ science.aad9416 PMID: 26989198</u>
- Harris K, Nielsen R. The genetic cost of neanderthal introgression. Genetics. 2016; 203(2):881–891. https://doi.org/10.1534/genetics.116.186890 PMID: 27038113
- Martin AR, Kanai M, Kamatani Y, Okada Y, Neale BM, Daly MJ. Clinical use of current polygenic risk scores may exacerbate health disparities. Nature Genetics. 2019; 51(4):584–591. <u>https://doi.org/10. 1038/s41588-019-0379-x PMID: 30926966</u>
- 66. Natri HM, Hudjashov G, Jacobs G, Kusuma P, Saag L, Darusallam CC, et al. Genetic architecture of gene regulation in Indonesian populations identifies QTLs associated with global and local ancestries. The American Journal of Human Genetics. 2022; 109(1):50–65. <u>https://doi.org/10.1016/j.ajhg.2021.11</u>. 017 PMID: 34919805
- Ward LD, Kellis M. Evidence of Abundant Purifying Selection in Humans for Recently Acquired Regulatory Functions. Science. 2012; 337(6102):1675–1678. <u>https://doi.org/10.1126/science.1225057</u> PMID: 22956687
- Naidoo T, Sjödin P, Schlebusch C, Jakobsson M. Patterns of variation in cis-regulatory regions: examining evidence of purifying selection. BMC Genomics. 2018; 19(1):95. https://doi.org/10.1186/s12864-017-4422-y PMID: 29373957
- Sabarís G, Laiker I, Preger-Ben Noon E, Frankel N. Actors with Multiple Roles: Pleiotropic Enhancers and the Paradigm of Enhancer Modularity. Trends in Genetics. 2019; 35(6):423–433. <u>https://doi.org/10.1016/j.tig.2019.03.006</u> PMID: 31005339
- 70. Findley AS, Zhang X, Boye C, Lin YL, Kalita CA, Barreiro L, et al. A signature of Neanderthal introgression on molecular mechanisms of environmental responses. PLOS Genetics. 2021; 17(9):e1009493. https://doi.org/10.1371/journal.pgen.1009493 PMID: 34570765
- Li Y, Banerjee S, Wang Y, Goldstein SA, Dong B, Gaughan C, et al. Activation of RNase L is dependent on OAS3 expression during infection with diverse human viruses. Proceedings of the National Academy of Sciences. 2016; 113(8):2241–2246. https://doi.org/10.1073/pnas.1519657113 PMID: 26858407

- Weiss CV, Harshman L, Inoue F, Fraser HB, Petrov DA, Ahituv N, et al. The cis-regulatory effects of modern human-specific variants. eLife. 2021; 10:e63713. <u>https://doi.org/10.7554/eLife.63713</u> PMID: 33885362
- **73.** Jagoda E, Xue JR, Reilly SK, Dannemann M, Racimo F, Huerta-Sanchez E, et al. Detection of Neanderthal Adaptively Introgressed Genetic Variants that Modulate Reporter Gene Expression in Human Immune Cells. Molecular Biology and Evolution. 2021; p. msab304.